

c/o Alvarez & Marsal Canada Inc. 400 Burrard Street Suite 1680, Commerce Place Vancouver, BC V6C 3A6 Ph: (604) 638-7440 Fax: (604) 638-7441

Sent by Email

May 31, 2016

Mackenzie Valley Land and Water Board, 7th Floor, 4922 48th St. PO Box 2130 Yellowknife, NT X1A 2P6 permits@mvlwb.com jmorse@mvlwb.com

Attention: Mr. Julian Morse

Re: Water License MV2015L2-0003 – Updated Care and Maintenance Plan Proponent Responses

NATCL is submitting responses to the reviewer comments provided on the Updated Care and Maintenance Plan submitted in March 2015.

Reviewer comments were considered by NATCL and responses have been provided within the table of comments spreadsheet. The table has been uploaded to the MVLWB ORS. Two additional documents have also been provided as per requests from the MVLWN regarding water quality data from underground.

If you have any questions or require further information, please contact myself at the Cantung Mine Site by telephone (604-759-0913 ext. 243), or by email at <u>enviro@natcl.ca</u>.

Yours truly, North American Tungsten Corporation

Kyle Conway, Sr. Environmental Coordinator

Attachments: Table of Comments, S4-13 WQ Data, S4-42 WQ Data

CC: B.Delaney, S. Sherwood, Cantung Enviro, C. Beveridge (Alvarez & Marsal)